Cognitive-behavioural therapy compared with standardised medical care for adults with dissociative non-epileptic seizures: the CODES RCT

Laura H Goldstein,^{1*} Emily J Robinson,^{2,3}
Izabela Pilecka,¹ Iain Perdue,¹ Iris Mosweu,⁴
Julie Read,¹ Harriet Jordan,¹ Matthew Wilkinson,^{5,6}
Gregg Rawlings,⁷ Sarah J Feehan,¹
Hannah Callaghan,⁸ Elana Day,¹ James Purnell,¹
Maria Baldellou Lopez,¹ Alice Brockington,⁹
Christine Burness,¹⁰ Norman A Poole,¹¹
Carole Eastwood,¹ Michele Moore,¹²
John DC Mellers,⁶ Jon Stone,⁸ Alan Carson,⁸
Nick Medford,⁶ Markus Reuber,¹³ Paul McCrone,⁴
Joanna Murray,¹⁴ Mark P Richardson,¹⁵
Sabine Landau^{2†} and Trudie Chalder^{16†}

¹Department of Psychology, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

²Department of Biostatistics and Health Informatics, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

³School of Population Health and Environmental Sciences, King's College London, London, UK

⁴King's Health Economics, King's College London, Institute of Psychiatry, Psychology and Neuroscience, London, UK

⁵Canterbury Christ Church University, Salamons Institute for Applied Psychology, Tunbridge Wells, UK

⁶South London and Maudsley NHS Foundation Trust, London, UK

⁷School of Clinical Psychology, University of Sheffield, Sheffield, UK

⁸Department of Clinical Neuroscience, Centre for Clinical Brain Sciences, University of Edinburgh, Edinburgh, UK

⁹Department of Neurology, Sheffield Teaching Hospitals NHS Foundation Trust, Sheffield, UK

¹⁰The Walton Centre NHS Trust, Liverpool, UK

¹¹Department of Neuropsychiatry, St George's Hospital, South West London and St George's NHS Mental Health NHS Trust, London, UK

¹²Centre for Social Justice and Global Responsibility, School of Law and Social Sciences, London South Bank University, London, UK

- ¹³Academic Neurology Unit, Royal Hallamshire Hospital, University of Sheffield, Sheffield, UK
- ¹⁴Department of Health Services & Population Research, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK
- ¹⁵Department of Basic and Clinical Neuroscience, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK
- ¹⁶Department of Psychological Medicine, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

*Corresponding author laura.goldstein@kcl.ac.uk †Joint last author

Declared competing interests of authors: Alan Carson reports being a paid editor of the *Journal of* Neurology, Neurosurgery and Psychiatry, and is the director of a research programme on functional neurological disorders; he gives independent testimony in court on a range of neuropsychiatric topics (50% pursuer, 50% defender). Sabine Landau is a paid editor of the Journal of Child Psychology and Psychiatry. Markus Reuber is the paid editor-in-chief of Seizure - European Journal of Epilepsy and receives authorship fees from Oxford University Press (Oxford, UK) in relation to a number of books about dissociative seizures. Markus Reuber benefited from the support of the National Institute for Health Research (NIHR) Sheffield Biomedical Research Centre (Translational Neuroscience). Mark P Richardson reports funding from Xenon Pharma (Burnaby, BC, Canada). Jon Stone reports independent expert testimony work for personal injury and medical negligence claims, royalties from UpToDate for articles on functional neurological disorders and runs a free non-profit self-help website, www.neurosymptoms.org. Jon Stone is also supported by an NHS Scotland NHS Research Scotland (NRS) Career Fellowship. Jon Stone and Alan Carson also acknowledge the financial support of NHS Research Scotland through the Edinburgh Clinical Research Facility. Laura Goldstein, Trudie Chalder and Sabine Landau report support from the NIHR Maudsley Biomedical Research Centre at the South London and Maudsley NHS Foundation Trust and King's College London. Emily J Robinson received salary support from the NIHR Biomedical Research Centre based at Guy's and St Thomas' NHS Foundation Trust and King's College London. Gregg Rawlings is entitled to authorship fees from Oxford University Press for several books on dissociative seizures.

Published June 2021 DOI: 10.3310/hta25430

Plain English summary

The CODES RCT

Health Technology Assessment 2021; Vol. 25: No. 43

DOI: 10.3310/hta25430

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Dissociative seizures resemble epileptic seizures or faints, but can be distinguished from them by trained doctors. Dissociation is the medical word for a 'trance-like' or 'switching off' state. People with dissociative seizures commonly have other psychological or physical problems. Quality of life may be low. The condition accounts for about one in every six patients seen in hospitals because of seizures.

We wanted to find out if people with dissociative seizures receiving standardised treatment would also benefit from a talking therapy, called cognitive-behavioural therapy, made specific to this disorder.

We did a randomised controlled trial to find out if people with dissociative seizures given standardised treatment and cognitive-behavioural therapy (talking therapy) would do better than those given standardised treatment alone.

Standardised treatment of dissociative seizures began with careful diagnosis from a neurologist and then further assessment and treatment from a psychiatrist.

In total, 368 people with dissociative seizures participated, with half receiving standardised treatment alone and half having talking therapy plus standardised treatment. We measured seizures and psychological and physical health in both trial groups. We also investigated whether or not cognitive-behavioural therapy was good value for money.

After 12 months, patients in both trial groups seemed to have fewer monthly seizures, but there was no advantage in the talking therapy group. Patients in the talking therapy group had more consecutive days without seizures, reporting less impact from them in everyday situations. Patients in the talking therapy group, and their doctors, considered improvements to be better, and patients in this group reported greater satisfaction with treatment. However, the talking therapy was expensive and not as cost-effective as hoped.

Interviews with patients and study clinicians showed that they valued aspects of both treatments and of the care provided by the multidisciplinary teams.

Overall, cognitive-behavioural therapy designed for dissociative seizures plus standardised treatment was not better at reducing the total numbers of seizures reported, but did produce several positive benefits for participants compared with standardised treatment alone.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.370

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 12/26/01. The contractual start date was in June 2014. The draft report began editorial review in April 2020 and was accepted for publication in August 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

Copyright © 2021 Goldstein et al. This work was produced by Goldstein et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk